Author's response to reviews

Title: Knockdown of anterior gradient 2 expression extenuates tumor-associated phenotypes of SNU-478 ampulla of Vater cancer cells

Authors:

Su Jin Kim (being97@hanmail.net)
Suyeon Jun (rktsnsanf@hanmail.net)
Hee-Yeon Cho (429iho@naver.com)
Dong Chul Lee (dc8840@hanmail.net)
Young Il Yeom (yeomyi@kribb.re.kr)
Jong Hyeok Kim (kih825@hallym.or.kr)
Dongchul Kang (dckang@hallym.ac.kr)

Version: 6
Date: 18 October 2014

Author's response to reviews:

(1) In response to the editorial request saying 'Please can you include a conclusion section, according to our guidelines', we have separated the 'Conclusion' section in the manuscript. We thank you for pointing out our mistake not to conform to the editorial guideline of the BMC Cancer. We apologize for this oversight on our part.

(2) In addition, we also have added a reference that substitutes for 'manuscript under revision' at line 417 in the 'Discussion' section because the results referred are published in 'Tohoku J Exp Med' recently.

Reference